Author's response to reviews

Title: Valvular involvement in ANCA-associated systemic vasculitis: a case report and literature review

Authors:

Chloe Lacoste (chloelacoste@yahoo.fr)
Nicolas Mansencal (nicolas.mansencal@apr.aphp.fr)
Mona Ben m'rad (mona.benmrad@gmail.com)
Catherine Goulon-Goeau (catherine.goulon-goeau@apr.aphp.fr)
Pascal Cohen (pascal.cohen@cch.aphp.fr)
Loic Guillemin (loic.guillemin@cch.aphp.fr)
Thomas Hanslik (thomas.hanslik@apr.aphp.fr)

Version: 3 Date: 21 February 2011

Author's response to reviews:

Dear Editors,

We have reviewed the manuscript "Valvular involvement in ANCA-associated systemic vasculitis: a case report and literature review" and made the major and minor revisions you suggested.

We have removed the highlighting and the capitalization in the names, suppressed the vertical lines from our tables, the merged cells that crossed rows and a few typographical errors.

All authors have read and approved the final manuscript.

Thank you for accepting this article for publication!

Sincerely,
Thomas Hanslik, MD, PhD.